Author's response to reviews

Title: Use of traditional complementary and alternative medicine for HIV patients in KwaZulu-Natal, South Africa

Authors:

Karl Peltzer (kpeltzer@hsrc.ac.za)
Natalie Friend-Dupreez (n.friend-dupreez@nhs.net)
Shandir Ramlagan (sramlagan@hsrc.ac.za)
Henry Fomundam (hfomundam@aol.com)

Version: 4 Date: 15 July 2008

Author's response to reviews:

the comments on the revised version, on the study limitations, has been included in the second revision